Author’s response to reviews

Title: Word Reading Threshold and Mild Cognitive Impairment: A Validation Study

Authors:

Genevieve Arsenault-Lapierre (genevieve.arsenault-lapierre@mail.mcgill.ca)
Howard Bergman (howard.bergman@mcgill.ca)
Howard Chertkow (howard.chertkow@mcgill.ca)

Version: 2 Date: 1 December 2011

Author’s response to reviews:

Dear Editors of BMC Geriatrics,

Enclosed you will find the revisions of our manuscript entitled: Word Reading Threshold and Mild Cognitive Impairment: A Validation Study.

We have corrected and modified the manuscript according to all the points raised by the reviewers. You will find enclosed the reviewers’ comments as well as our answers to comments.